Author’s response to reviews

Title: The role of abdominal aortic calcification in predicting mortality in hemodialysis patients: a retrospective observational study

Authors:

Daqing Hong (daqing.hong@gmail.com)
Shukun Wu (wushukun1980@163.com)
Lei Pu (lei-pu@163.com)
Fang Wang (wqqf1121@vip.sina.com)
Junru Wang (junrugood@163.com)
Zhengtong Wang (wangzhengtong@gmail.com)
Hui Gao (kellygh@163.com)
Yue Zhang (z_amao@126.com)
Fei Deng (deng_fei@sohu.com)
Guisen Li (guisenli@163.com)
Qiang He (heqiang@medmail.com.cn)
Li Wang (scwangli62@163.com)

Version: 2 Date: 18 March 2013

Author’s response to reviews: see over
The role of abdominal aortic calcification in predicting mortality in hemodialysis patients: a retrospective observational study

Daqing HONG1,§, Shukun WU1, §, Lei PU1, Fang WANG1, Junru WANG1, Zhengtong WANG2, Hui GAO1, Yue ZHANG1, Fei DENG1, Guisen LI1, Qiang HE1, Li WANG1.

1 Division of Nephrology, Sichuan Academy of Medical Sciences and Sichuan Provincial People’s Hospital, Chengdu 610072, China
2 Division of Nephrology, Jinhua Municipal Center Hospital, Jinhua 321000, China
§ Both authors contributed equally to this work.

Running title: Vascular, aortic calcification & mortality

Address correspondence to:
Prof. Li WANG,
Division of Nephrology,
Sichuan Academy of Medical Sciences and Sichuan Provincial People’s Hospital,
Chengdu 610072, China
Email: scwanglt62@163.com
Telephone: (86)028-87038358
Fax: (86)028-87038528

Reasons to support the publication of this article in BMC Nephrology: This article meets the scope of publication of this journal. We performed radiographs of the abdomen, pelvis and hands to study the impact of different areas of vascular calcification (VC) upon mortality in patients undergoing hemodialysis. We found that abdominal aortic calcification alone was not a good predictor of patient mortality in our population, which differed from most previous studies, but a strong association was found between mortality and concurrent VC in the abdomen, pelvis and hands. We concluded that the presence of VC in lateral abdominal radiographs may not be superior to other radiographs for predicting patient outcome.
Response to editorial Request:

1. Ethical Approval
   In the Methods section, please include the full name of the ethical committee that granted approval, listing a reference number if obtained.

   I have finished this part in the Methods section.

2. Copy-editing

   After reading through your manuscript, we feel that the quality of written English needs to be improved before the manuscript can be considered further.

   Please also ensure that your revised manuscript conforms to the journal style

   These requests have been done.